Author's response to reviews

Title: Postprandial fullness correlates with rapid inflow of gastric content into duodenum but not with chronic gastritis.

Authors:

Nobutake Yamamichi (nyamamic- tky@umin.ac.jp)
Takeshi Shimamoto (tsimamoto@gmail.com)
Chihiro Minatsuki (furukawa-ymn@umin.ac.jp)
Yoko Yoshida (yoshida.yoko@nsn.co.jp)
Mitsuhiro Fujishiro (mtfujish-kkr@umin.ac.jp)
Shinya Kodashima (kodashima-tky@umin.ac.jp)
Jun Kato (katojun@wakayama-med.ac.jp)
Osamu Goto (ogotou-gi@umin.ac.jp)
Satoshi Ono (satoshi-tky@umin.ac.jp)
Keiko Niimi (k-niimi@umin.ac.jp)
Yu Takahashi (yutakahashi-tky@umin.ac.jp)
Maki Konno-Shimizu (maki11shimizu@yahoo.co.jp)
Masao Ichinose (ichinose@wakayama-med.ac.jp)
Kazuhiko Koike (kkoike-tky@umin.ac.jp)

Version: 5 Date: 18 December 2011

Author's response to reviews:

We added the sentence in the "Authors' contributions" paragraph according to the below-mentioned indication (Major revision). We also excluded the the title from two image files (Figure 1 and Figure 2), according to the below-mentioned indication (Minor revision).

Thank you for your consideration.

Yours sincerely,

Nobutake Yamamichi, M.D., Ph.D.

-----------------------

Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number.